# UC Berkeley

UC Berkeley Previously Published Works

Title

Temporal genomic contrasts reveal rapid evolutionary responses in an alpine mammal during recent climate change

Permalink https://escholarship.org/uc/item/7wr476fd

Journal PLOS Genetics, 15(5)

ISSN

1553-7390

Authors

Bi, Ke Linderoth, Tyler Singhal, Sonal <u>et al.</u>

Publication Date

2019

DOI 10.1371/journal.pgen.1008119

Peer reviewed



# G OPEN ACCESS

**Citation:** Bi K, Linderoth T, Singhal S, Vanderpool D, Patton JL, Nielsen R, et al. (2019) Temporal genomic contrasts reveal rapid evolutionary responses in an alpine mammal during recent climate change. PLoS Genet 15(5): e1008119. https://doi.org/10.1371/journal.pgen.1008119

Editor: Alex Buerkle, University of Wyoming, UNITED STATES

Received: December 29, 2018

Accepted: April 1, 2019

Published: May 3, 2019

**Copyright:** © 2019 Bi et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Raw sequence data have been deposited on the NCBI Sequence Read Archive under BioProject Accession IDs PRJNA312223 (SRA: SRP070198) and PRJNA312234 (SRA: SRP191784). Associated numerical data underlying the main results and figures have been deposited on figshare (doi:10. 6084/m9.figshare.7871912). Custom scripts developed for data filtering and population genetic analysis have been archived in GitHub (https:// github.com/tplinderoth/ngsQC doi: 10.5281/ **RESEARCH ARTICLE** 

# Temporal genomic contrasts reveal rapid evolutionary responses in an alpine mammal during recent climate change

Ke Bi<sup>1,2©</sup>, Tyler Linderoth<sup>1,3©</sup>, Sonal Singhal<sup>1,3</sup>, Dan Vanderpool<sup>4</sup>, James L. Patton<sup>1,3</sup>, Rasmus Nielsen<sup>3</sup>, Craig Moritz<sup>1,3,5</sup>, Jeffrey M. Good<sup>6,4,6</sup>\*

Museum of Vertebrate Zoology, University of California, Berkeley, California, United States of America,
Computational Genomics Resource Laboratory (CGRL), California Institute for Quantitative Biosciences (QB3), University of California, Berkeley, California, United States of America,
Department of Integrative Biology, University of California, Berkeley, California, United States of America,
Department of Biological Sciences, University of Montana, Missoula, Montana, United States of America,
Research School of Biology and Centre for Biodiversity Analysis, Australian National University, Canberra, ACT, Australia,
Wildlife Biology Program, University of Montana, Missoula, MT, United States of America

• These authors contributed equally to this work.

\* jeffrey.good@umontana.edu

# Abstract

Many species have experienced dramatic changes in their abundance and distribution during recent climate change, but it is often unclear whether such ecological responses are accompanied by evolutionary change. We used targeted exon sequencing of 294 museum specimens (160 historic, 134 modern) to generate independent temporal genomic contrasts spanning a century of climate change (1911-2012) for two co-distributed chipmunk species: an endemic alpine specialist (Tamias alpinus) undergoing severe range contraction and a stable mid-elevation species (T. speciosus). Using a novel analytical approach, we reconstructed the demographic histories of these populations and tested for evidence of recent positive directional selection. Only the retracting species showed substantial population genetic fragmentation through time and this was coupled with positive selection and substantial shifts in allele frequencies at a gene, Alox15, involved in regulation of inflammation and response to hypoxia. However, these rapid population and gene-level responses were not detected in an analogous temporal contrast from another area where T. alpinus has also undergone severe range contraction. Collectively, these results highlight that evolutionary responses may be variable and context dependent across populations, even when they show seemingly synchronous ecological shifts. Our results demonstrate that temporal genomic contrasts can be used to detect very recent evolutionary responses within and among contemporary populations, even in the face of complex demographic changes. Given the wealth of specimens archived in natural history museums, comparative analyses of temporal population genomic data have the potential to improve our understanding of recent and ongoing evolutionary responses to rapidly changing environments.

#### zenodo.2592044, https://github.com/tplinderoth/ ABCutils doi: 10.5281/zenodo.2591546).

Funding: This research was supported by an NSERC postdoctoral fellowship (KB). University of California Berkeley VCR-BiGCB, and the Gordon and Betty Moore Foundation (GBMF2983). TL was supported in part by the NIH Genomics Training Grant (Grant T32HG000047-13). Fieldwork was supported by grants from the Yosemite Conservancy, the National Park Service Inventory and Monitoring Program, the National Geographic Society (Grant 8190-07), and the National Science Foundation (DEB 0640859). Additional instrumentation, laboratory, and computational support was provided by the University of Montana Genomics Core, supported by a grant from the M. J. Murdock Charitable Trust, the Vincent J. Coates Genomics Sequencing Laboratory at UC Berkeley, supported by NIH S10 Instrumentation Grants S10RR029668 and S10RR027303, and the Texas Advanced Computing Center (TACC) at the University of Texas at Austin. Grants from the Eunice Kennedy Shriver National Institute of Child Health and Human Development (R01HD073439; JMG) and the National Institute of General Medical Sciences (R01GM098536; JMG) supported the development of experimental protocols for exome capture in mice that were utilized in the current study. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Museum specimens represent an irreplaceable archive that can be used to understand how species respond to rapid environmental change. We recovered genomic data from archived samples spanning a century of climate change in co-distributed declining versus stable species of montane chipmunks. Applying novel statistical methods, we find evidence for strong positive selection on a physiologically relevant gene despite increased population fragmentation in the declining species. Our results reveal rapid evolutionary responses, but also highlight that genetic adaptation has been insufficient to prevent range collapse in this endemic alpine species. These findings illustrate how biological archives can be used to pinpoint genetic responses through time to better understand how species are responding to rapidly changing environments.

# Introduction

Rapid environmental change threatens global biodiversity and has led to population declines in many species [1–5]. Although phenotypic plasticity may enable populations to track rapidly changing climates, evolutionary adaptation will often be essential for long-term persistence [6]. Disentangling plasticity from evolutionary responses ultimately requires resolving the genetic basis of adaptation. However, it remains challenging to differentiate recent or ongoing positive selection from stochastic genetic drift in contemporary populations that are also undergoing extreme demographic changes [7, 8]. Natural history museum collections may hold the key to overcoming many of these difficulties by providing crucial temporal information on species distributions, phenotypes, and population genetic variation spanning periods of recent environmental change [9–11]. Temporal genomic contrasts have begun to yield powerful insights into recent evolutionary responses in humans [12, 13] and other species [14–17], indicating that genetic analyses of biological archives will be an effective tool for understanding evolutionary responses to rapid anthropogenic climate change [18].

Using contrasts between early 20<sup>th</sup> century and modern museum surveys, Moritz and colleagues [1] showed that the ranges of several high elevation small mammal species in the Yosemite National Park (YNP) region of the Sierra Nevada mountains (California, USA) have retracted upward over the past century. This and associated studies [19, 20] demonstrated the potential of using museum archives to understand species and community-level ecological responses during periods of recent climate change. Contemporary range shifts towards higher latitudes, elevations, or both have now been documented in many terrestrial species [5, 19, 21, 22], and are generally thought to reflect direct and indirect population responses to warming temperatures [22, 23]. However, these works have also highlighted that closely related species can differ markedly in the magnitude and direction of their ecological responses, for reasons that are often not clear [1, 20]. Currently, we know very little about how recent range shifts have affected evolutionary processes within species, or the extent to which evolutionary genetic responses have been synchronous within and between co-distributed species.

Here we focus on two chipmunk species within the YNP montane mammal community that show different ecological and phenotypic responses over the last century of climate change (Fig 1). Western chipmunks have long been considered models for niche partitioning by elevation and habitat type [24–26]. The alpine chipmunk (*Tamias alpinus*) is an ecological specialist endemic to the high elevation alpine habitats of the Sierra Nevada Mountains. The lodgepole chipmunk (*T. speciosus*) occurs more broadly across mid- to high-elevation subalpine coniferous forests of California. *Tamias alpinus* has undergone severe range contraction driven by



**Fig 1. Historic and modern sampling localities.** *Tamias speciosus* and *T. alpinus* specimens were collected from Yosemite National Park (**A**, **B**) and the Southern Sierras (**C**). Historical sampling localities (1911–1916) are shown in filled blue circles and modern (2003–2012) in filled red crosses. The vertical bar plots in the right panel show the elevation (kilometers) range of the species over the past century with sequenced samples indicated by colored dots

PLOS GENETICS

within the bars. Upper elevation limits were determined for *T. speciosus* (shown in gray), but sequenced samples were selected from mid-elevation sites where sufficient historical and modern samples were available. The trapping resurvey was not designed to ascertain upper limits for *T. alpinus* from either transect area. Sequencing sample sizes (n) are indicated under the elevation bars. While *T. speciosus* maintained a stable range, *T. alpinus* populations have severely contracted upwards in elevation (indicated with arrows) throughout their distribution [20]. The inset (bottom right) USA map shows the ranges of *T. alpinus* (dark gray) and *T. speciosus* (gray) with the state of California with Yosemite National Park (YNP, northern) and the Southern Sierras (SS, southern) study areas are indicated with arrows and outlined in white.

#### https://doi.org/10.1371/journal.pgen.1008119.g001

extirpation of lower elevation populations [1, 20] combined with pronounced shifts in diet and cranial morphology [27] across the alpine zone of YNP and elsewhere in the Sierra Nevada mountains. Spatial modeling of current versus historical ranges across YNP suggests that the strong upward contraction of T. alpinus is best explained by increases in minimum winter temperatures; competing models including co-occurrence with other species of chipmunks or changes in the distribution of preferred vegetation types did not improve prediction of the observed range contraction in this species [28, 29]. That increasing minimum temperature alone had the strongest impact makes sense in that there has been little change in vegetation across the high elevation talus slopes preferred by this species. By contrast, range contraction in a mid-elevation chipmunk species, T. senex, is best explained by changes in its preferred vegetation types [28, 29]. For YNP T. alpinus, there is also evidence of strong directional selection on skull morphology over the past century [30], whereas the range, diet, and morphology of the partially overlapping lodgepole chipmunk (T. speciosus) has remained relatively stable within YNP [20, 27, 31]. As is generally the case [32], it remains unclear why the montane specialist, T. alpinus, has contracted during the past century whereas its more widely-distributed congener, T. speciosus, has not.

These two species set up a natural contrast, allowing us to understand how differing ecological responses during periods of rapid environmental change correspond to differing evolutionary responses. A previous temporal survey of eight microsatellite markers revealed increased subdivision and declining allelic diversity in YNP *T. alpinus*, but no significant changes in overall population genetic variation of *T. speciosus* over the same interval [33]. Synthesizing descriptions of phenotypic [30], behavioral [31], and genetic variation [33] into a detailed understanding of demographic and evolutionary responses in these species requires genomic data. Towards this end, Bi et al. [9] used a custom exon capture platform to enrich and sequence ~11,000 exons (~4 Megabases or Mb across 6,249 protein-coding genes) from 20 early 20<sup>th</sup> century and 20 modern YNP *T. alpinus*. These genome-wide data confirmed signatures of increasing population subdivision in this retracting species and illustrated the potential for targeted genomic experiments to generate high quality data from archived specimens. However, neither of these preceding analyses had sufficient sampling to determine whether evident increases in population structure were due to reductions in local population size, migration rates, or both.

Here, we build on these previous works [9, 33] by generating ~9.4 Mb of targeted exome sequence data from 303 chipmunks (194 *T. alpinus*, 100 *T. speciosus*, and 9 samples from 4 other species). We used these comparative population genomic data to quantify evolutionary responses over the past century at two scales (Fig 1). First, we sequenced 96 modern (48 *T. alpinus* and 48 *T. speciosus* collected between 2003–2008) and 108 historic (56 *T. alpinus*, 52 *T. speciosus* collected in 1915–1916) samples collected across geographic transects in YNP for the focal species. Second, we generated an independent geographic transect of 38 modern (2003–2012) and 52 historic (1911–1916) *T. alpinus* in the Southern Sierras (SS), where this species also shows range contraction [20], to test to what extent evolutionary responses across the

range of this declining alpine specialist have been consistent. To analyze these temporal data, we developed a novel analytical framework based on Approximate Bayesian Computation (ABC) that allowed us to quantify changes in population sizes and migration rates in the context of demographic history, and then to localize positive selection on standing genetic variation at specific genes. Our results provide new insights into recent evolutionary responses in this system and demonstrate how a genomic time-series approach can be broadly applied to other archived specimens to improve understanding of evolutionary responses over the past few centuries.

## **Results and discussion**

#### Targeted enrichment of genomic data spanning a century of climate change

High sequencing coverage is necessary to reliably genotype ancient DNA samples [34] due to extensive DNA degradation [35]. This persistent technical challenge makes whole genome resequencing of historic mammalian populations impractical, especially in species without high-quality reference genomes (e.g., *Tamias*). Therefore, we designed a custom targeted capture to enrich and sequence exons from over 10,107 protein-coding genes (9.4 Megabases or Mb total) in 294 *T. alpinus* and *T. speciosus* samples (S1 Table) and nine samples from four other chipmunk species (total n = 303). We sampled geographic transects of modern and historic (~100 year-old) populations in YNP for both species as well as an independent SS transect of *T. alpinus*, where this species has also contracted [20]. An average of 49 individuals were sequenced per population (Fig 1). This design allowed us to (i) compare stable and retracting species within the same montane mammal community (YNP), and to (ii) determine to what extent the same evolutionary responses have occurred across two transects (YNP and SS) spanning the latitudinal range of the range-retracted species *T. alpinus*.

Exome enrichment was highly specific (90-93% of cleaned reads on target) and sensitive (>92% of the target regions sequenced), resulting in high coverage of targeted regions (26– 35× average individual coverage per population, <u>S2 Table</u>). Although historic DNA samples are notorious for poor technical performance, all targeted T. speciosus and T. alpinus individuals yielded moderate to high coverage data with similar capture success between modern and historic samples. Analysis of mitochondrial DNA indicated that empirical error rates were ~fourfold higher in historic (0.16%) versus modern samples (0.04%), due primarily to DNA damage typical of century-old museum samples [9, 36]. Although nucleotide misincorporations associated with deamination of methylated cytosine bases (C-to-T and G-to-A) were most common near the ends of DNA fragments, such changes remained elevated throughout the sequence (S1 Fig). Therefore, we applied several quality filters to remove all putative misincorporations and to mitigate other common sources of genotyping error [9] (S3 Table). All filters were uniformly applied to historic, modern, and simulated data (see below) to facilitate comparisons across time periods and species. After filtering, we identified 20,395, 10,395, and 10,954 high-quality single nucleotide polymorphisms (SNPs) in YNP T. speciosus, YNP T. alpinus, and SS T. alpinus, respectively.

#### Population genetics and demographic inference

Upwards range contraction in montane environments can lead to increased population structure and reduced genetic diversity due to decreased population size. We observed a consistent trend towards relatively minor reductions in nucleotide diversity in modern versus historic samples of *T. alpinus* and *T. speciosus* estimated at two spatial scales: metapopulations (e.g., YNP or SS) and demes of spatially clustered sampling localities ( $\theta_{\pi}$  and  $\theta_{W}$ ; S2 Fig, S4 Table), consistent with prior results [33]. We then quantified the degree of population genetic structure within each species by estimating the global fixation index ( $F_{ST}$ ) for historic and modern populations. Within YNP, population structure was relatively low overall but increased nearly two-fold in modern *T. alpinus* ( $F_{ST,historic} = 0.032$ ,  $F_{ST,modern} = 0.058$ ), consistent with increased population fragmentation as this species has contracted upwards [33]. By contrast, population structure in *T. speciosus* increased only slightly over time ( $F_{ST,historic} =$ 0.027,  $F_{ST,modern} = 0.030$ ). We detected very little uncertainty in our estimates of the site frequency spectrum (SFS), and estimates of global  $F_{ST}$  showed non-overlapping 0.95 bootstrapped confidences intervals for all pairwise temporal contrasts (S4 Table). Collectively, these patterns suggest that the stronger increase in  $F_{ST}$  observed for YNP *T. alpinus* is not simply explained by reductions in per deme nucleotide diversity, which are of similar magnitude across YNP and SS *T. alpinus* and YNP *T. speciosus* (S2 Fig).

To further evaluate changes in the spatial patterning of population genetic structure, we used NGSadmix to conduct a maximum likelihood (ML) analysis of historic and modern population structures [37]. We detected a substantial increase from two to six genetic clusters for YNP *T. alpinus*, compared to stable overall population structure (K = 2 across both time points) within *T. speciosus* (Fig 2A; S5 Table). Principal component analyses of these data also indicate less genetic similarity among modern YNP *T. alpinus* samples (Fig 2B). These genome-wide estimates incorporate genotype uncertainty to provide an accurate overview of changes in genetic diversity and structure over recent timescales. Although consistent in the direction of change, the increases in YNP *T. alpinus* population genetic structure appear even more striking than previously detected using lower resolution data [33, 38]. By contrast, there was minimal structure in *T. speciosus* within the same ecosystem whether considering a few microsatellite loci [33] or thousands of SNPs (Fig 2A).

Consistent with patterns of increased population genetic structure, spatial models of occurrence indicate that range contraction has reduced local connectivity between suitable habitat patches for YNP *T. alpinus* [33]. Given that range contraction has also been detected at the southern limit of the range of *T. alpinus* [20], we next tested to what extent similar temporal signatures of increased genetic structure were apparent in SS populations (Fig 1). Population structure in the SS transect has also increased ( $F_{ST,historic} = 0.034$ ,  $F_{ST,modern} = 0.044$ ), but to a lesser extent than observed in YNP, and with no overall increase in the number of distinct genetic groups (modern and historic K = 2; Fig 2A). This suggests some variation in the local genetic consequences of seemingly synchronous, range-wide contractions. However, we note that our power to detect changes in overall population structure may have been limited by the fact that historic and modern sampling localities were more spatially clustered in SS when compared to YNP (Fig 1).

Rapid range shifts could also increase the likelihood of hybridization if there are changes in the degree of sympatry or in relative densities among closely related species [39, 40]. Occasional hybridization appears to be common in western chipmunks [41], including evidence for ancient mitochondrial introgression from the broadly distributed least chipmunk, *T. minimus*, into *T. speciosus* [42]. The alpine chipmunk is very closely related to *T. minimus* with evidence of historical gene flow [38], raising the possibility that recent range fluctuations have induced hybridization between *T. alpinus* and either *T. minimus* or *T. speciosus*. However, we found no evidence for recent appreciable nuclear gene flow between *T. alpinus* and adjacent populations of *T. speciosus* or with neighboring (lower elevation) populations of *T. minimus* (S3 Fig). Thus, recent range collapse does not appear to have led to the breakdown of reproductive barriers between this high elevation endemic and other co-distributed species [38]. We did, however, detect nuclear introgression from *T. speciosus* into at least one *T. minimus* sample (S3 Fig), suggesting that reproductive isolation remains incomplete between these species despite strong ecological partitioning [26].



**Fig 2. Temporal and spatial population genetic structure. (A)** Genetic clustering by general sampling locality (left, pie charts) and individual (bar graph to the right of each map) across historic and modern samples based on ngsAdmix analyses. Each individual is partitioned into colored segments that indicate cluster membership. Pie charts represent the sum of all individuals' membership in each cluster at each general locality on the map. Proximate individual sample localities were pooled for clarity following Rubidge and colleagues [33]. The inferred best number of clusters (K) is shown on the top of each bar graph. Global F<sub>ST</sub> estimates between localities are indicated at the bottom of each map. **(B)** Each data point in the PCA plot represents an individual specimen. The first and second two principal components (PCs) explain 3.7% and 2.6% of the total genetic variance.

https://doi.org/10.1371/journal.pgen.1008119.g002

These basic descriptions of genetic variation expand the scope and resolution of previous analyses of a few microsatellite loci genotyped in *T. alpinus*, *T. minimus*, and *T. speciosus* [33, 38], and from more limited exon capture data from *T. alpinus* [9]. However, disentangling changes in migration versus local effective population size, and identifying genes under positive selection in the context of recent demographic change, requires more comprehensive analyses. Therefore, we developed a novel analytical framework to fully exploit our temporal dataset. Traditional population genetic analyses often assume even sampling across space and time, yet studies using museum specimens are typically imbalanced because of limited availability of samples. ABC is well suited for demographic inference under such circumstances because biased temporal sampling and sample processing can be simulated for populations

that have experienced complex demographic histories. Accordingly, simulations can be filtered in the same way as observed data (e.g., removal of SNPs associated with errors in historic DNA), permitting meaningful comparisons between expected and observed results. A major difficulty of ABC is the choice of statistics that sufficiently describe demographic parameters of interest. Multiple, jointly informative, summary statistics are often used to sufficiently estimate parameters while reducing the risk of any particular statistic biasing the results [43]. The site frequency spectrum (SFS) is often an optimal choice for fitting demographic histories since many commonly used summary statistics can be derived from it. In practice, high dimensionality and low count categories of joint site frequency spectra make them difficult to fit. Consequently, we developed an effective means of fitting binned two-dimensional SFS (2D-SFS) using an ABC framework designed to infer population histories from serially sampled metapopulations (S4 Fig).

We constructed 2D-SFS for each of the three pairwise temporal contrasts by pooling individuals across sampling localities within YNP or SS into a single metapopulation per time period (S5 Fig). While allele frequencies should be highly correlated over such short time scales, the overall shape of the joint spectrum should change in predictable ways in response to various population-level processes. We fitted multiple demographic models to the 2D-SFS (S6 Fig; S7 Fig) describing population size (constant size, bottlenecks, expansions) and connectivity (migration) among subpopulations or demes representing spatially clustered localities (see S6-S9 Tables and S1 Text for details on model selection, evaluation, and inference). The best fitting population history for YNP T. alpinus was characterized by relatively small but constant deme effective sizes through time (~1,350 individuals) and an approximately threefold decrease in migration within the past ~90 years (Fig 3). This fitted demographic model fits well with our observation of increased population structure in YNP T. alpinus (Fig 2) despite only minor reductions in nucleotide diversity per deme through time (S2 Fig, S4 Table). In contrast, both SS T. alpinus and YNP T. speciosus were found to have much larger effective deme sizes (~4,600 and ~4,560 individuals respectively) and higher migration rates overall. Consistent with the observed increase in  $F_{ST}$ , SS T. alpinus showed some evidence for a recent, very small decline in effective size (Fig 3). YNP T. speciosus was the only population that showed a clear signature of size change, albeit related to a historic (pre 20<sup>th</sup> C) population expansion (Fig 3).

We found that modern samples tended overall to be less genetically similar to each other than did historic samples in all three comparisons (Fig 2B; S8 Fig). Consistent with this observation, the fitted demographic histories for each species and transect support a recent (< 90 years ago) decrease in migration among demes (Fig 3; see S1 Text, S6 Table). Decreased migration is expected if climate change is broadly affecting the amount of connectivity between suitable habitat patches available to species within this montane community [33]. However, as noted above, substantially increased genetic structuring was most evident in YNP *T. alpinus* (Fig 2A). Long-term changes in gene flow should ultimately affect the genetic composition of metapopulations across these landscapes. It is possible that higher overall migration rates and larger effective population sizes have so far buffered the population genetic effects in *T. speciosus* and SS *T. alpinus*. However, genetic structure could accumulate over time according to the inferred histories.

These results emphasize the utility of high-resolution demographic inference from genomic data not only for reconstructing population histories, but also as a potentially powerful conservation management tool [18, 44, 45]. An ABC framework is generalizable to other temporally sampled genetic datasets, allowing high-resolution inference into demographic histories over shallow evolutionary timescales that are relevant to recent anthropogenic climate change. In species of conservation concern, signatures of reduced migration could be used to motivate





#### (modern sample)

**Fig 3. Population histories inferred with Approximate Bayesian Computation.** Population histories fitted with ABC showing the general topology and posterior median demographic parameter values averaged across the best fitting models for YNP *T. speciosus*, YNP *T. alpinus*, and SS *T. alpinus*, respectively. The fitted parameters are the modern effective number of individuals per deme (Ne), migration rates (m), intrinsic growth rates (r), and the timing of demographic events (1 generation = 1 year). History widths are proportional to the deme effective sizes though time. The four-deme depiction (YNP *T. alpinus* and *T. speciosus*) represents histories fitted with an island model having equal pairwise migration rates between all demes, and where the actual number of islands equaled the number of sampled demes. Three SS *T. alpinus* demes were sampled and modeled as islands with migration rates being allowed to vary between different pairs of demes. The events with dashed lines for SS *T. alpinus* are relatively uncertain (see <u>S1 Text</u>), but if any demographic changes did occur they are as shown (note that a bottleneck is likely to have been weak and the size is not to scale).

https://doi.org/10.1371/journal.pgen.1008119.g003

introductions between populations prior to significant genetic erosion, buffering against the future loss of genetic diversity and the accumulation of deleterious variation [46]. The benefits of such proactive efforts would have to be weighed carefully relative to the potential risks of introducing locally maladaptive variation [47].

#### Targets of positive directional selection

Connections between broad ecological patterns and the genetic structure of populations are often intuitive and predictable. Upwards range contraction in *T. alpinus* is associated with reduced connectivity between suitable montane habitats [33], which we infer has reduced migration between patches and increased genetic drift. However, *a priori* expectations for patterns of recent adaptive evolution are far less predictable in these species. Recent range shifts [20] and temporal changes in diet and skull morphologies [27, 30] are both consistent with stronger directional selection gradients in YNP and SS *T. alpinus* relative to *T. speciosus*. On the other hand, lower effective population sizes and reduced migration (at least in YNP) should make selection relatively less effective in *T. alpinus*. Likewise, more effective adaptive responses could explain why larger and more connected populations of *T. speciosus* have remained stable in the face of common environmental stressors.

To begin to tease these issues apart, we tested for specific genetic changes that might underlie recent adaptive responses in these species by directly comparing genetic differences between historic and modern populations. All three temporal population pairs are very closely related (Fig 2), however, they are also separated by changes in population structure and sizes (Fig 3) that may confound standard signatures of positive selection [48]. Therefore, we tested



**Fig 4. Derived alleles showing significant frequency shifts between historic and modern populations of** *Tamias alpinus* in YNP. (A) Five outlier SNPs (a-e, FDR q < 0.001) are labeled on a plot of the neutral per site temporal  $F_{ST}$  distribution (modern versus historic) estimated by OutFLANK. The histogram of observed  $F_{ST}$  (yellow bins) is shown with the inferred neutral distribution (blue line). (B) Unfolded two-dimensional site frequency spectrum (2D-SFS) for SNPs between historic (x-axis) and modern (y-axis) YNP *Tamias alpinus* specimens. The color of each data point represents the number of SNPs (depicted by the color key) belonging to that particular 2D-SFS category. Arrows point to the five outliers (a-e) showing the only significant allele frequency shifts over time. (C) Derived allele frequencies of the five outliers SNPs plotted against sample elevation. Individual sample localities were pooled into 100-meter elevational bands to enable allele frequency estimation. (D) The position of the five outliers mapped onto the *Alox15* gene in the *Mus musculus* reference genome. SNP 'a' is a synonymous mutation (A/G) in exon 2 (Chromosome 11:70350801); SNP 'b' is synonymous (T/C) and maps to exon 8 (11:70347260); SNPs 'c' (T/A), 'd' (T/G), and 'e' (T/G) are located in the intron between exon 8 and 9.

https://doi.org/10.1371/journal.pgen.1008119.g004

for individual SNPs that had undergone large frequency shifts between historic and modern populations using an approach that is robust to the confounding influence of complex population histories on the genomic distribution of  $F_{ST}$  [49–51]. We found no significant allele frequency shifts over time in YNP *T. speciosus* or SS *T. alpinus*. In contrast, we identified five outlier SNPs in YNP *T. alpinus* populations (false discovery rate [FDR] q-value < 0.01) relative to the inferred null distribution of per-site, genome-wide  $F_{ST}$  between the temporally sampled populations (genome-wide temporal  $F_{ST} = 0.012$ ; Fig 4A). To verify the inference of positive selection on these SNPs, we compared the observed  $F_{ST}$  values to null distributions simulated under the best ABC-fitted demographic history for YNP *T. alpinus* (Fig 3). Our simulated  $F_{ST}$ distributions were in close agreement with the overall observed  $F_{ST}$  values. Thus, it is very unlikely that demography alone could produce the extreme changes in allele frequencies that we observed at the outlier loci (p-value < 3e-7; S9 Fig).

Derived allele frequencies at all five differentiated SNP positions increased ~threefold in the modern populations (average frequencies of 0.22 historic versus 0.65 modern; Fig 4B) and all were located in the protein-coding gene, Arachidonate 15-Lipoxygenase (*Alox15*) (Fig 4D). *Alox15* is a broadly expressed lipoxygenase involved in the resolution of acute inflammation through the generation of lipid-derived signaling molecules known as resolvins [52–54]. *Alox15* expression has been associated with cardiovascular disease, oxidative stress, and

response to hypoxia [55–57] as part of the Hypoxia-inducible factor-1 $\alpha$  (HIF-1 $\alpha$ ) regulation pathway [58]. Two of the outliers represent synonymous changes in non-adjacent exons (positions a, b; Fig 4D) while the three other SNPs (positions c-e) were at non-coding positions within the same intron. All five positions were in strong linkage disequilibrium (historic r<sup>2</sup> = 0.86; modern r<sup>2</sup> = 0.93) in YNP *T. alpinus* but invariant in all other populations except for one site (b) that was at similar frequency across the SS *T. alpinus* temporal contrast (historic = 0.13, modern = 0.2).

Given an ~500m contraction of the low elevation range limit in YNP *T. alpinus* over the last century [20], the temporal allele frequency shifts at *Alox15* could simply reflect non-sampling of extinct low elevation populations. To test this, we first estimated *Alox15* allele frequencies as a function of elevation by pooling individuals into discrete 100-meter elevation bands. We observed the largest increases in derived allele frequencies at low to mid-elevation localities (Fig 4C), and mean derived allele frequencies were not significantly correlated with elevation in either of the temporal samples (historical  $R^2 = 0.46$ , p-value = 0.14; modern  $R^2 = 0.34$ , p-value = 0.42). Furthermore, all five positions remained strong outliers in our temporal F<sub>ST</sub> contrasts when we excluded low elevation sampling localities that were present only in the historic YNP *T. alpinus* transect (OutFLANK FDR q < 0.05, ABC-fitted F<sub>ST</sub> distribution p-value = 4e-7). Thus, evolutionary responses at *Alox15* are consistent with *in situ* evolutionary change primarily among remnant demes below the upper bound of the modern YNP *T. alpinus* range (<3200 meters elevation).

In principle, the large shift in *Alox15* allele frequencies observed between historic and modern samples could be driven by changes in habitats, food availability, or some other non-climate related environmental factor. However, based on previous modeling of changes in the elevational range [28] and the function of *Alox15*, we suggest that physiological response to warming is the strongest current hypothesis. Winter temperature appears to be a primary limiting factor in the distribution of *T. alpinus*, with range contractions strongly tracking upslope shifts in minimum winter temperatures [28]. Increases in minimum winter temperatures at mid-elevations are resulting in reduced YNP snowpacks [59, 60], which Rubidge and colleagues suggested might reduce over-winter survival of *T. alpinus* through loss of critical thermal insulation of hibernacula [28]. Interestingly, arousal from hibernation has been shown to induce oxidative stress and hypoxia [61] and *Alox15* shows strong seasonal induction in other species of hibernating squirrels [62].

A potential link between the intensity of selection on variation at *Alox15* and changes in winter snowpack could also explain why we did not detect selection at the same gene in the SS transect. *Tamias alpinus* populations in the Southern Sierra are fixed for ancestral alleles at all but one of the outlier YNP SNPs, suggesting that these populations may lack genetic variation at *Alox15* that is putatively adaptive in YNP. Moreover, SS *T. alpinus* populations are currently found above ~3200 meters—above the elevation range showing the largest allele frequency shifts in YNP—and overall snowpack has been more stable in the southern Sierra during the last century [59]. Though speculative, these scenarios help illustrate how evolutionary responses among populations may depend on both adaptive potential (i.e., standing genetic variation) and local environmental conditions.

#### Conclusions

Temporal sampling of genomic data has the potential to provide powerful insights into the evolutionary effects of rapid environmental change [18]. Here we built on previous works [1, 9, 33] by generating targeted genome-wide sequence data from 294 chipmunks spanning a century of climate change. By integrating high throughput sequencing, cost and time-effective

targeted enrichment technologies, and sophisticated inference methods, we provide powerful comparative insights into demographic and evolutionary responses of two montane species experiencing rapid environmental change. Our genomic time-series approach demonstrates one way that historical archives can be used to study biological responses during recent environmental change [9, 11, 18]. Temporal genomic data can provide a means to understand the current state of populations and their potential evolutionary trajectories, providing powerful tools to inform the conservation of populations experiencing changing environments.

The identification of targets of positive selection during the recent upslope range contraction in *T. alpinus* points to a candidate gene and potential phenotypes associated with physiological stress that warrant further study. We caution that further evidence, such as differences in over-winter survival across genotypes or other functional studies, are necessary to demonstrate a causal relationship between *Alox15* and response to climatic-induced stress. Further, our capture experiment only covered a subset of protein coding genes (~50%) and did not include extensive coverage of regulatory regions that may often modulate rapid evolutionary responses [63]. That said, alpine chipmunks also show greater stress response to changes in external conditions [64], a narrower range of activity patterns [31], and more pronounced shifts in diet and functional aspects of cranial morphology when compared to *T. speciosus* over the past century [27, 30]. Thus, the combination of phenotypic, behavioral, and now genetic evidence points to some component of physiological stress as a key factor in the greater sensitivity of *T. alpinus* to environmental change.

Even in the absence of links to specific phenotypes or fitness, the identification of evolutionary responses at specific genes should help inform future on-ground studies focused on identifying the proximate causes of warming-related population declines across the range of this or other affected species [32]. Indeed, the potential for adaptive evolution to rescue populations in decline has emerged as an important concept in conservation biology [65], with increasing efforts to directly incorporate evolutionary principles into conservation planning [66]. As a cautionary note, our results suggest that putatively adaptive responses in *T. alpinus* at *Alox15* (Fig 4), as well as rapid shifts in functional morphology and diet [27, 30], have nonetheless been insufficient to prevent extensive extirpation of lower elevation populations of this alpine specialist.

Comparative analyses of species range shifts over the past century have provided powerful insights into the ecological impacts of and biological responses to rapid environmental changes [1, 19, 21–23]. Here we have begun to extend these ideas to a comparative population genomic framework. Moving forward, we suggest that the true power of analyzing genomes and phenotypes of historical museum archives lies in the potential to extend across species [9, 11]. Though the occurrence of museum records tend to be highly punctuated through space and time for a given species, historic collection efforts, such as those led by Joseph Grinnell and other early naturalists, usually surveyed many co-distributed species. With comparable contemporary sampling efforts, these invaluable archives will enable comparative community level insights into the impacts of and evolutionary responses to rapidly changing environments.

#### Methods

#### Ethics statement

All animals sampled in the modern era were collected following procedures approved by the University of California, Berkeley Animal Care and Use Committee (Permit number R278–0315). Permits were provided by Yosemite National Park and Sequoia-Kings Canyon National Park.

### **Biological samples**

Tamias speciosus and T. alpinus surveyed in this study were collected from montane transects in Yosemite National Park (YNP) and the Southern Sierras (SS). Historic samples were collected by Joseph Grinnell and his colleagues from 1911 to 1916, and are preserved as dried skins in the Museum of Vertebrate Zoology (MVZ), at the University of California, Berkeley. Modern samples were collected from the same sites by the 'Grinnell Resurvey' team led by MVZ researchers and collaborators from 2003 to 2012 (Fig 1; S1 Table). We examined 100 YNP T. speciosus (52 historic, 48 modern), 104 YNP T. alpinus (56 historic, 48 modern), and 90 SS T. alpinus (52 historic, 38 modern) from each transect. We also sampled six T. minimus (the Least chipmunk) collected east of YNP, which were used to test for potential hybridization between T. alpinus and T. minimus [38]. Furthermore, we included one sample each of three other species (T. striatus, T. ruficaudus, and T. amoenus) in order to polarize SNPs identified in our focal populations. Historic DNA was extracted from toe pad tissue (~3 x 3 mm) in a separate dedicated laboratory using a previously described protocol [9]. DNA was extracted from modern samples using Qiagen DNeasy Blood and Tissue kits following the manufacturer's protocol. Genomic libraries for all samples were constructed following Meyer and Kircher [67] with slight modifications [9].

#### Exome capture design and implementation

We used RNA-seq [68] to sequence and assemble [69] transcriptomes for multiple tissues sampled from a single modern SS *T. alpinus* to serve as a reference for exome capture probe design. We targeted exonic regions (6.9 Mb, including flanking introns and intergenic regions) corresponding to 8,053 *T. alpinus* genes targeted by our previous array-based capture experiments in chipmunks [9, 70, 71]. In addition, we extracted a broad set of candidate genes from the AmiGO and NCBI protein databases with functional annotations that were potentially relevant to environmental stress responses (e.g., HSP/HSF, hemoglobin, cytokines, apoptosis, immunity, oxidative stress, oxidative phosphorylation). We then used a BLASTx search against these genes to locate 2,054 orthologous transcripts (2.4 Mb) from the *Tamias* transcriptome and included these transcripts in our capture. We also targeted the complete mitochondrial genome (~16.4 Kb) to assess empirical error rates and five previously sequenced nuclear genes [42, 72] to use as positive controls in post-capture qPCR assays of global enrichment efficiency. Probes were designed and manufactured by NimbleGen (SeqCap EZ Developer kits).

Barcoded genomic libraries were pooled together and hybridized in seven independent reactions with *Tamias* Cot-1 DNA and barcode-specific blocking oligonucleotides. Six hybridization experiments were used for the focal species (one per time point for each of the three temporal contrasts) and one additional capture was performed on pooled libraries from six *T. minimus* and three outgroup samples (*T. striatus, T. ruficaudus,* and *T. amoenus*). After hybridization, each of the enriched genomic libraries were amplified using PCR and sequenced using one lane of Illumina HiSeq2000 per capture (100-bp paired-end).

#### Data processing

Bioinformatic processing of exon capture data followed our previous protocols [9, 70]. All raw sequencing reads were treated to remove adapters, exact duplicates, low complexity (i.e., runs of ambiguous or mononucleotide sequence), and reads sourced from bacteria and human contamination. Overlapping paired reads were merged to avoid inflated estimates of coverage and biased genotype likelihoods. We used filtered sequencing reads (28.9 Gb) from 48 modern YNP *T. alpinus* samples to generate *de novo* assemblies with ABySS [73] that were then merged using Blat [74], CD-HIT [75], and CAP3 [76] to remove redundancies. This total assembly was

then compared to the original targets to construct a non-redundant target reference of 21,128 assembled contiguous sequences (contigs) totaling 20.8 Mb, and error-corrected following Bi and colleagues [9]. We then aligned cleaned reads from *T. alpinus*, *T. speciosus*, and *T. mini-mus* samples to the *T. alpinus* reference using Novoalign (http://www.novocraft.com). Nucleotide positions were filtered at individual, contiguous sequence, and position levels of quality control following our previously described methods [9] (S3 Table) using the script snpCleaner (https://github.com/tplinderoth/ngsQC/tree/master/snpCleaner). For each of the three temporal transects, we retained the intersection of filtered contigs between all historic and modern populations. As a result, 2,569, 2,451, and 2,738 contigs (11.6–13% of the total) were eliminated from YNP *T. speciosus*, YNP *T. alpinus*, and SS *T. alpinus* datasets, respectively. At the site level, we removed sites showing unusually high or low coverage, excessive strand bias, end distance bias, base quality bias, and map quality bias. We also filtered out sites with extensive missing data among samples within each population.

We were particularly attentive to errors associated with long-term DNA degradation. Postmortem nucleotide damage from hydrolytic deamination causes conversion from cytosine (C) to uracil (U) residues resulting in misincorporation of thymine (T) during PCR amplification [34, 35, 77]. We conservatively removed all C-to-T and G-to-A (i.e., the reverse complement of the C-to-T change with respect to the original PCR template molecules) SNP positions from the datasets to avoid inaccurate population genetic inferences stemming from base misincorporation. In total, 9.0, 9.3, and 8.5 Mb of data from YNP *T. speciosus*, YNP *T. alpinus*, and SS *T. alpinus* passed all quality controls and were used in subsequent analyses.

#### **Population genetic analyses**

We used probabilistic methods for variant discovery and allele frequency estimation as implemented within ANGSD [78]. Using a population-specific SFS estimated from allele frequency likelihoods as a prior, we obtained allele frequency posterior probabilities and called SNPs using a 0.95 probability cutoff of being variable. The realSFS function was used to generate 1,000 bootstrap replicates of the folded site frequency spectrum (SFS) for each metapopulation by resampling per site allele frequency likelihoods. We then used ANGSD to estimate the number of segregating sites (S), Watterson's theta ( $\theta_W$ ), pairwise nucleotide diversity ( $\theta\pi$ ), and Tajima's D in the historic and modern T. alpinus and T. speciosus metapopulations. For each metapopulation, we generated 100 estimates of  $\theta_{\rm W}$  and  $\theta\pi$  using randomly chosen SFS bootstrap replicates as priors to evaluate sensitivity of these point estimates on the SFS prior. Additionally we estimated diversity statistics for demes within metapopulations, the former representing spatially clustered sampling localities. For each transect, population differentiation within and between the modern and historic populations was determined using probabilistic methods for estimating F<sub>ST</sub> [79] and individual covariance matrices for principal component analysis (PCA) as implemented in ngsTools [80]. Confidence intervals (0.95) for global F<sub>ST</sub> were generated from 1,000 bootstrap replicates of per-site F<sub>ST</sub> values. To compare allele frequencies over time, we estimated the 2D-SFS between the pooled modern and pooled historic demes of each transect (i.e., three 2D-SFS comparisons). SNPs identified in T. speciosus and T. alpinus were polarized relative to T. striatus, T. ruficaudus, and T. amoenus. We further examined population genetic structure using NGSadmix [37], which estimates admixture proportions from genotype likelihoods. We ran 10 replicates for K (number of clusters) ranging from 1–10 and summarized results (S5 Table) across runs to determine the best K [81]. To test for hybridization between T. alpinus and T. minimus samples, we used the program ADMIXTURE [82] to estimate individual ancestries using one randomly sampled SNP per contig.

Next we developed an ABC framework for fitting binned 2D-SFS from serially sampled populations or metapopulations (S4 Fig) and used this approach to test hypotheses about the demographic histories of the sampled chipmunk populations. Additional details on demographic model construction, simulations, model selection, and inference are provided in S1 Text. Briefly, we fitted 5-9 explicit demographic models (S6 Fig) characterized by possible changes in migration and population size to each of the temporal contrasts. We performed 25,000 simulations per model, drawing parameter values from uniform or log-uniform prior distributions and then simulating ~20.2 Mb of sequence data for each individual under the specified history using the coalescent simulator fastsimcoal [83]. Lineages from the different demes were sampled at the present (modern sample) and 90 generations in the past (historic sample) according to the actual number of sampled individuals. Then all samples within a respective time period were pooled and the historic versus modern 2D-SFS was calculated. A custom script was used to calculate diagonal and off diagonal bins of the joint SFS (S4 Fig), which served as our ABC summary statistic. We used the R package 'abc' [84] to calculate model posterior probabilities and to evaluate the reliability of our model selection procedure. We considered the best fitting models for each species/transect to be those with the highest posterior probabilities (S7 Table) and we used a cross validation procedure to determine error rates associated with model choice (S8 Table). To aid model choice, we also considered the fit of the maximum likelihood (ML) estimate for each model to our observed data. We evaluated goodness-of-fit for the selected models by comparing the Euclidean distance between our observed and simulated 2D-SFS bins (S9 Table).

We considered SNPs with large allele frequency shifts between the modern and historic time periods that could not be attributed to demography as evidence for positive selection. We used the program OutFLANK [50] to detect  $F_{ST}$  outlier SNPs (FDR q-value < 0.01), empirically adjusting the degrees of freedom of  $\chi^2$ -distributed  $F_{ST}$  values to account for the influence of demography. We then compared the observed SNP  $F_{ST}$  values to null exome-wide and persite  $F_{ST}$  distributions generated by performing 1,500 neutral simulations under the best fitting population history for YNP *T. alpinus*.

# **Supporting information**

**S1 Text. Approximate Bayesian computation inference.** (PDF)

**S1 Fig. Patterns of base misincorporations in historic samples.** The frequencies of the 12 types of substitutions (y-axis) are plotted as a function of distance from the 5' and 3'-ends of the DNA molecules (x-axis). The first 50 bp of the reads are shown. The substitution frequency of each particular type is calculated as the proportion of a particular alternative (non-reference) base type at a given site along the read, and is coded in different colors and line patterns as indicated at the top of the plots: "X-> Y" indicates a change from reference base type X to alternative base type Y.

(PDF)

**S2** Fig. Per deme estimates of nucleotide diversity for historical and modern samples of *T*. *speciosus* and *T. alpinus*. Estimates represented by a single point reflect demes that were only sampled at a single timepoint. (PDF)

**S3 Fig. Genetic ancestry composition of three chipmunk species.** Each individual specimen is represented by a vertical bar partitioned into colored segments indicating their proportion of ancestry from each species. Results are shown for modern samples. (PDF)

S4 Fig. Demographic inference procedure. Diagram of the demographic inference method based on fitting bins of the 2D-SFS with Approximate Bayesian Computation (ABC) that was used to infer the *Tamias* population histories. The ABC summary statistics calculated in step 3 are the off-diagonal (left 2D-SFS) and diagonal (right 2D-SFS) bins of the 2D-SFS, where each bin's value is the sum of the counts contained within the bin. The bin width refers to the number of 2D-SFS categories on either side of the diagonal/off-diagonal that are included in the bin and determines the amount of resolution (finer bins for higher resolution) and noise dampening (wider bins) when fitting the 2D-SFS. We used a bin width of 2 in the current study. For assessing model fit in step 5,  $D_{ML,obs}$  is the Euclidean distance between the observed or pseudo observed 2D-SFS bins and bins from the maximum likelihood (ML) history under the chosen model. Pseudo observed values are a set of bins under the ML history that are treated like the observed bins.



**S5 Fig. Unfolded and folded two-dimensional site frequency spectrum (2D-SFS) for SNPs between historic (x-axis) and modern (y-axis) specimens.** The color of each data point represents the number of SNPs belonging to that particular category in the 2D-SFS, which is depicted in the color key inset. (PDF)

S6 Fig. Demographic models fit with ABC. The general demographic models that were fit with ABC included histories with demes bottlenecking into the present (A, B, G), historic population expansion (C, D, J), historic population expansion followed by a bottleneck (H), and constant deme size (E, F, N). The shaded circles within the history outlines represent different demes. The actual number of simulated demes equaled the number of sampling localities, which were 10 for YNP T. alpinus, 3 for SS T. alpinus, and 8 for YNP T. speciosus. Demes were simulated under an island model (symmetric migration and identical deme sizes). Per generation migration rates (m) among all deme pairs were equal for the YNP T. alpinus and T. speciosus contrasts, while for SS T. alpinus the migration rates between different pairs of demes were allowed to differ. Solid versus dotted arrows between demes represent potentially different migration rates, while no arrows represent no migration between demes. The parameter m<sub>hist</sub> is the historic migration rate between two demes, m<sub>mod</sub> is the modern migration rate between a pair of demes, t<sub>mig change</sub> is the number of generations in the past at which the migration rate changes, rgrow is the intrinsic growth rate for population expansion, tgrow\_stop is the number of generations in the past that expansion stops, rshrink is the intrinsic growth rate of population decline,  $t_{shrink}$  is the number of generations in the past that a bottleneck starts, and 'mod  $N_e$ ' is the effective size of each deme at the present. The presence of '90' in the models involving migration rate change and/or a population bottleneck indicates that these demographic changes were set to occur 90 generations prior to the modern sample. (PDF)

**S7 Fig. Assessment of model fit for inferred** *Tamias* **chipmunk demographic histories. (A)** Comparison of the empirical cumulative distribution functions (ECDF) for the Euclidean distance between the observed and expected 2D-SFS bins, d(ML,obs), for all YNP *T. alpinus* (YNPA), SS *T. alpinus* (SSA), and YNP *T. speciosus* (YNPS) maximum likelihood histories. For each model, 1,000 simulations under its ML history were performed to generate the expected joint frequency spectra from which the distribution of d(ML,obs) was calculated. Histories with left-shifted ECDF curves are more likely to resemble the true demographic history. (**B-D**) Comparison of D<sub>ML,obs</sub> and D<sub>ML,pseudo</sub> distributions for the best-fitting histories for YNP *T. alpinus* (**B**), SS *T. alpinus* (**C**), and YNP *T. speciosus* (**D**).  $D_{ML,obs}$  is the Euclidean distance between the observed and expected joint SFS bins under a model's ML history, while  $D_{ML,pseudo}$  is the distance between a single set of joint SFS bins under the ML history (pseudo observed) and the expected joint SFS bins. For each ML history, 1,000 simulations were performed to generate the expected joint spectra. The values within the distributions are Weitzman's coefficient of overlapping (OVL), ranging from 0 to 1, which quantifies the area of overlap of the two distributions. More overlap between the two distributions indicates that the ML history is more likely to represent the true demography. (PDF)

**S8 Fig. Principal Component Analysis (PCA) plots based on genetic covariance among individuals.** The first 3 principal components (PCs) are shown in **(A-C)**. Each point in the PCA plot represents an individual specimen. The proportion of the genetic variance explained by the first 10 PCs is shown in **(D-F)**. (PDF)

**S9 Fig. Expected**  $F_{ST}$  **distribution for YNP** *T. alpinus* **under histories inferred with ABC.** The expected distribution of neutral  $F_{ST}$  per site and exome-wide under the best fitting demographic histories for YNP *T. alpinus* inferred with ABC. Each  $F_{ST}$  distribution was generated from 1,500 simulations under the maximum likelihood histories for the best fitting demographic models for YNP *T. alpinus*, B, F, H, and N. The positions of the five observed  $F_{ST}$  outlier SNPs (see Fig 4) are plotted onto the expected per site  $F_{ST}$  distributions. (PDF)

**S1 Table.** Historic and modern sample information. (XLSX)

**S2 Table. Exon capture data production information.** (XLSX)

**S3 Table. Individual, contig, and site-level quality filters used for genotyping.** (PDF)

S4 Table. Population genetic summary statistics calculated with ANGSD and ngsTools on quality-filtered data.

(PDF)

**S5 Table. Population genetic structure inferred with ngsAdmix.** (XLSX)

S6 Table. ABC demographic parameter posterior and maximum likelihood estimates under the best fitting models for each chipmunk temporal contrast. (XLSX)

S7 Table. Demographic model posterior probabilities for three *Tamias* temporal contrasts approximated using the rejection method at a tolerance level of 0.8%. (PDF)

**S8** Table. Demographic model confusion matrix for the three temporal contrasts generated using cross validation with rejection sampling tolerance set at 0.8%. (XLSX)

S9 Table. Pairwise comparisons between  $D_{ML,obs}$  empirical CDFs for the different models fit to the three temporal contrasts in terms of Kolmogorov-Smirnov 2-Sample test

statistics (blue-labeled rows) and their associated p-values (red-labeled rows). (XLSX)

## Acknowledgments

The authors would like to thank Chris Conroy, Eileen Lacey, Karen Rowe, Kevin Rowe, and Jack Sullivan for providing access to samples, A.-P. Assis, L. Chow, P. Moore, and C. Patton for and field assistance, Sara Keeble for lab support at the University of Montana, Tali Hammond for identification of candidate genes, and Maria Santos for assistance with maps. We thank the Evolutionary Genetics Lab (EGL) and its manager, Lydia Smith, for providing help and support throughout this study and members of the University of Montana UNVEIL network for helpful discussion.

#### **Author Contributions**

Conceptualization: Ke Bi, Tyler Linderoth, Rasmus Nielsen, Craig Moritz, Jeffrey M. Good.

Data curation: Ke Bi, Tyler Linderoth.

Formal analysis: Ke Bi, Tyler Linderoth.

Funding acquisition: Craig Moritz, Jeffrey M. Good.

Investigation: Ke Bi, Tyler Linderoth, Dan Vanderpool.

Methodology: Ke Bi, Tyler Linderoth, Dan Vanderpool, Rasmus Nielsen, Jeffrey M. Good.

Project administration: Craig Moritz, Jeffrey M. Good.

Resources: Sonal Singhal, James L. Patton, Rasmus Nielsen, Craig Moritz, Jeffrey M. Good.

Software: Tyler Linderoth, Sonal Singhal.

Supervision: Rasmus Nielsen, Craig Moritz, Jeffrey M. Good.

Validation: Ke Bi, Tyler Linderoth.

Visualization: Ke Bi, Tyler Linderoth.

Writing - original draft: Ke Bi, Tyler Linderoth, Craig Moritz, Jeffrey M. Good.

Writing – review & editing: Ke Bi, Tyler Linderoth, Sonal Singhal, Rasmus Nielsen, Craig Moritz, Jeffrey M. Good.

#### References

- Moritz C, Patton J, Conroy C, Parra J, White G, Beissinger S. Impact of a century of climate change on small-mammal communities in Yosemite National Park, USA. Science. 2008; 322:261–4. <u>https://doi.org/10.1126/science.1163428</u> PMID: 18845755
- Sinervo B, Mendez-de-la-Cruz F, Miles DB, Heulin B, Bastiaans E, Villagran-Santa Cruz M, et al. Erosion of lizard diversity by climate change and altered thermal niches. Science. 2010; 328:894–9. <a href="https://doi.org/10.1126/science.1184695">https://doi.org/10.1126/science.1184695</a> PMID: 20466932
- Urban MC. Accelerating extinction risk from climate change. Science. 2015; 348:571–3. https://doi.org/ 10.1126/science.aaa4984 PMID: 25931559
- 4. Pecl GT, Araujo MB, Bell JD, Blanchard J, Bonebrake TC, Chen IC, et al. Biodiversity redistribution under climate change: Impacts on ecosystems and human well-being. Science. 2017; 355:eaai9214. https://doi.org/10.1126/science.aai9214 PMID: 28360268
- Freeman BG, Scholer MN, Ruiz-Gutierrez V, Fitzpatrick JW. Climate change causes upslope shifts and mountaintop extirpations in a tropical bird community. Proceedings of the National Academy of Sciences of the United States of America. 2018; 115:11982–7. https://doi.org/10.1073/pnas.1804224115 PMID: 30373825

- Hoffmann AA, Sgro CM. Climate change and evolutionary adaptation. Nature. 2011; 470:479–85. https://doi.org/10.1038/nature09670 PMID: 21350480
- Gienapp P, Teplitsky C, Alho JS, Mills JA, Merila J. Climate change and evolution: disentangling environmental and genetic responses. Molecular Ecology. 2008; 17:167–78. https://doi.org/10.1111/j.1365-294X.2007.03413.x PMID: 18173499
- Merilae J, Hendry AP. Climate change, adaptation, and phenotypic plasticity: the problem and the evidence. Evolutionary Applications. 2014; 7:1–14. https://doi.org/10.1111/eva.12137 PMID: 24454544
- Bi K, Linderoth T, Vanderpool D, Good JM, Nielsen R, Moritz C. Unlocking the vault: next-generation museum population genomics. Molecular Ecology. 2013; 22:6018–32. https://doi.org/10.1111/mec. 12516 PMID: 24118668
- Fountain T, Nieminen M, Siren J, Wong SC, Hanski I. Predictable allele frequency changes due to habitat fragmentation in the Glanville fritillary butterfly. Proceedings of the National Academy of Sciences of the United States of America. 2016; 113:2678–83. <u>https://doi.org/10.1073/pnas.1600951113</u> PMID: 26903642
- Holmes M, Hammond T, Wogan G, Walsh R, LaBarbera K, Wommack E, et al. Natural history collections as windows on evolutionary processes. Molecular Ecology. 2016; 25:864–81. https://doi.org/10. 1111/mec.13529 PMID: 26757135
- Mathieson I, Lazaridis I, Rohland N, Mallick S, Patterson N, Roodenberg SA, et al. Genome-wide patterns of selection in 230 ancient Eurasians. Nature. 2015; 528:499–503. <u>https://doi.org/10.1038/</u> nature16152 PMID: 26595274
- Fu Q, Posth C, Hajdinjak M, Petr M, Mallick S, Fernandes D, et al. The genetic history of Ice Age Europe. Nature. 2016; 534:200–5. https://doi.org/10.1038/nature17993 PMID: 27135931
- Mikheyev AS, Tin MM, Arora J, Seeley TD. Museum samples reveal rapid evolution by wild honey bees exposed to a novel parasite. Nature Communications. 2015; 6:7991. <u>https://doi.org/10.1038/</u> ncomms8991 PMID: 26246313
- Campbell-Staton SC, Cheviron ZA, Rochette N, Catchen J, Losos JB, Edwards SV. Winter storms drive rapid phenotypic, regulatory, and genomic shifts in the green anole lizard. Science. 2017; 357:495–7. https://doi.org/10.1126/science.aam5512 PMID: 28774927
- Nunziata SO, Lance SL, Scott DE, Lemmon EM, Weisrock DW. Genomic data detect corresponding signatures of population size change on an ecological time scale in two salamander species. Molecular Ecology. 2017; 26:1060–74. https://doi.org/10.1111/mec.13988 PMID: 28026889
- Cridland JM, Ramirez SR, Dean CA, Sciligo A, Tsutsui ND. Genome sequencing of museum specimens reveals rapid changes in the genetic composition of honey bees in California. Genome Biology and Evolution. 2018; 10:458–72. https://doi.org/10.1093/gbe/evy007 PMID: 29346588
- Diez-Del-Molino D, Sanchez-Barreiro F, Barnes I, Gilbert MTP, Dalen L. Quantifying temporal genomic erosion in endangered species. Trends in Ecology & Evolution. 2018; 33:176–85.
- Tingley MW, Beissinger SR. Cryptic loss of montane avian richness and high community turnover over 100 years. Ecology. 2013; 94:598–609. PMID: 23687886
- Rowe KC, Rowe KMC, Tingley MW, Koo MS, Patton JL, Conroy CJ, et al. Spatially heterogeneous impact of climate change on small mammals of montane California. Proceedings of the Royal Society B-Biological Sciences. 2015; 282:20141857.
- Parmesan C, Ryrholm N, Stefanescu C, Hill JK, Thomas CD, Descimon H, et al. Poleward shifts in geographical ranges of butterfly species associated with regional warming. Nature. 1999; 399:579–83.
- 22. Chen IC, Hill JK, Ohlemueller R, Roy DB, Thomas CD. Rapid range shifts of species associated with high levels of climate warming. Science. 2011; 333:1024–6. https://doi.org/10.1126/science.1206432 PMID: 21852500
- Freeman BG, Lee-Yaw JA, Sunday JM, Hargreaves AL. Expanding, shifting and shrinking: The impact of global warming on species' elevational distributions. Global Ecology and Biogeography. 2018; 27:1268–76.
- 24. Grinnell J, Storer T. Animal Life in Yosemite. Berkeley, CA: University of California Press; 1924.
- **25.** Brown JH. Mechanisms of competitive exclusion between two species of chipmunks. Ecology. 1971; 52:305–11.
- **26.** Heller HC. Altitudinal zonation of chipmunks (*Eutamias*): interspecific aggression. Ecology. 1971; 52:312–9.
- Walsh R, Assis A, Patton J, Marroig G, Dwason T, Lacey E. Morphological and dietary responses of chipmunks to a century of climate change. Global Change Biology. 2016; 22:3233–52. https://doi.org/ 10.1111/gcb.13216 PMID: 26732228

- 28. Rubidge EM, Monahan WB, Parra JL, Cameron SE, Brashares JS. The role of climate, habitat, and species co-occurrence as drivers of change in small mammal distributions over the past century. Global Change Biology. 2011; 17:696–708.
- 29. Santos MJ, Thorne JH, Moritz C. Synchronicity in elevation range shifts among small mammals and vegetation over the last century is stronger for omnivores. Ecography. 2015; 38:556–68.
- Assis APA, Patton JL, Hubbe A, Marroig G. Directional selection effects on patterns of phenotypic (co) variation in wild populations. Proceedings of the Royal Society B-Biological Sciences. 2016; 283:20161615.
- Hammond TT, Palme R, Lacey EA. Ecological specialization, variability in activity patterns and response to environmental change. Biology Letters. 2018; 14:20180115. <u>https://doi.org/10.1098/rsbl.</u> 2018.0115 PMID: 29950317
- Cahill AE, Aiello-Lammens ME, Fisher-Reid MC, Hua X, Karanewsky CJ, Ryu HY, et al. How does climate change cause extinction? Proceedings of the Royal Society B-Biological Sciences. 2013; 280:20121890.
- Rubidge EM, Patton JL, Lim M, Burton AC, Brashares JS, Moritz C. Climate-induced range contraction drives genetic erosion in an alpine mammal. Nature Climate Change. 2012; 2:285–8.
- Axelsson E, Willerslev E, Gilbert MTP, Nielsen R. The effect of ancient DNA damage on inferences of demographic histories. Molecular Biology and Evolution. 2008; 25:2181–7. <u>https://doi.org/10.1093/</u> molbev/msn163 PMID: 18653730
- Briggs AW, Stenzel U, Johnson PLF, Green RE, Kelso J, Prufer K, et al. Patterns of damage in genomic DNA sequences from a Neandertal. Proceedings of the National Academy of Sciences of the United States of America. 2007; 104:14616–21. https://doi.org/10.1073/pnas.0704665104 PMID: 17715061
- Sawyer S, Krause J, Guschanski K, Savolainen V, Pääbo S. Temporal patterns of nucleotide misincorporations and DNA fragmentation in ancient DNA. PloS ONE. 2012; 7:e34131. <u>https://doi.org/10.1371/journal.pone.0034131</u> PMID: 22479540
- Skotte L, Korneliussen TS, Albrechtsen A. Estimating individual admixture proportions from next generation sequencing data. Genetics. 2013; 195:693–702. https://doi.org/10.1534/genetics.113.154138 PMID: 24026093
- Rubidge EM, Patton JL, Moritz C. Diversification of the Alpine Chipmunk, *Tamias alpinus*, an alpine endemic of the Sierra Nevada, California. BMC Evolutionary Biology. 2014; 14:34. <u>https://doi.org/10. 1186/1471-2148-14-34 PMID: 24559294</u>
- Kelly B, Whiteley A, Tallmon D. The Arctic melting pot. Nature. 2010; 468:891. <u>https://doi.org/10.1038/468891a</u> PMID: 21164461
- Chunco AJ. Hybridization in a warmer world. Ecology and Evolution. 2014; 4:2019–31. https://doi.org/ 10.1002/ece3.1052 PMID: 24963394
- Sullivan J, Demboski JR, Bell KC, Hird S, Sarver B, Reid N, et al. Divergence with gene flow within the recent chipmunk radiation (*Tamias*). Heredity. 2014; 113:185–94. <u>https://doi.org/10.1038/hdy.2014.27</u> PMID: 24781803
- Reid N, Demboski JR, Sullivan J. Phylogeny estimation of the radiation of western North American chipmunks (*Tamias*) in the face of introgression using reproductive protein genes. Systematic Biology. 2012; 61:44–62. https://doi.org/10.1093/sysbio/syr094 PMID: 21878471
- Beaumont MA. Approximate Bayesian computation in evolution and ecology. Annual Review of Ecology, Evolution, and Systematics. 2010; 41:379–406.
- Orlando L, Cooper A. Using ancient DNA to understand evolutionary and ecological processes. Annual Review of Ecology Evolution and Systematics. 2014; 45:573–98.
- Beichman AC, Huerta-Sanchez E, Lohmueller KE. Using genomic data to infer historic population dynamics of nonmodel organisms. Annual Review of Ecology, Evolution, and Systematics. 2018; 49:433–56.
- **46.** Whiteley AR, Fitzpatrick SW, Funk WC, Tallmon DA. Genetic rescue to the rescue. Trends in Ecology & Evolution. 2015; 30:42–9.
- Aitken SN, Whitlock MC. Assisted gene flow to facilitate local adaptation to climate change. Annual Review of Ecology, Evolution, and Systematics. 2013; 44:367–88.
- Nielsen R. Molecular signatures of natural selection. Annual Review of Genetics. 2005; 39:197–218. https://doi.org/10.1146/annurev.genet.39.073003.112420 PMID: 16285858
- Lotterhos KE, Whitlock MC. Evaluation of demographic history and neutral parameterization on the performance of FST outlier tests. Molecular Ecology. 2014; 23:2178–92. <u>https://doi.org/10.1111/mec.</u> 12725 PMID: 24655127

- Whitlock MC, Lotterhos KE. Reliable detection of loci responsible for local adaptation: inference of a null model through trimming the distribution of F<sub>ST</sub>. American Naturalist. 2015; 186:S24–S36. <u>https://doi.org/10.1086/682949</u> PMID: 26656214
- Hoban S, Kelley JL, Lotterhos KE, Antolin MF, Bradburd G, Lowry DB, et al. Finding the genomic basis of local adaptation: pitfalls, practical solutions, and future directions. American Naturalist. 2016; 188:379–97. https://doi.org/10.1086/688018 PMID: 27622873
- Spite M, Norling LV, Summers L, Yang R, Cooper D, Petasis NA, et al. Resolvin D2 is a potent regulator of leukocytes and controls microbial sepsis. Nature. 2009; 461:1287–91. <u>https://doi.org/10.1038/</u> nature08541 PMID: 19865173
- Serhan CN. Pro-resolving lipid mediators are leads for resolution physiology. Nature. 2014; 510:92– 101. https://doi.org/10.1038/nature13479 PMID: 24899309
- 54. Kim SN, Akindehin S, Kwon HJ, Son YH, Saha A, Jung YS, et al. Anti-inflammatory role of 15-lipoxygenase contributes to the maintenance of skin integrity in mice. Scientific Reports. 2018; 8:8856. https:// doi.org/10.1038/s41598-018-27221-7 PMID: 29891910
- Kayama Y, Minamino T, Toko H, Sakamoto M, Shimizu I, Takahashi H, et al. Cardiac 12/15 lipoxygenase-induced inflammation is involved in heart failure. Journal of Experimental Medicine. 2009; 206:1565–74. https://doi.org/10.1084/jem.20082596 PMID: 19546247
- Mashima R, Okuyama T. The role of lipoxygenases in pathophysiology; new insights and future perspectives. Redox Biology. 2015; 6:297–310. <u>https://doi.org/10.1016/j.redox.2015.08.006</u> PMID: 26298204
- 57. Lundqvist A, Sandstedt M, Sandstedt J, Wickelgren R, Hansson GI, Jeppsson A, et al. The Arachidonate 15-Lipoxygenase enzyme product 15-HETE is present in heart tissue from patients with ischemic heart disease and enhances clot formation. PLoS ONE. 2016; 11:e0161629. https://doi.org/10.1371/ journal.pone.0161629 PMID: 27552229
- Yao L, Nie X, Shi S, Song S, Hao X, Li S, et al. Reciprocal regulation of HIF-1 alpha and 15-LO/15-HETE promotes anti-apoptosis process in pulmonary artery smooth muscle cells during hypoxia. Prostaglandins & Other Lipid Mediators. 2012; 99:96–106.
- Howat IM, Tulaczyk S. Trends in spring snowpack over a half-century of climate warming in California, USA. Annals of Glaciology. 2005; 40:151–6.
- Mote PW, Hamlet AF, Clark MP, Lettenmaier DP. Declining mountain snowpack in western north America. Bulletin of the American Meteorological Society. 2005; 86:39–50.
- Carey HV, Frank CL, Seifert JP. Hibernation induces oxidative stress and activation of NF-kB in ground squirrel intestine. Journal of Comparative Physiology B-Biochemical Systemic and Environmental Physiology. 2000; 170:551–9.
- Orr AL, Lohse LA, Drew KL, Hermes-Lima M. Physiological oxidative stress after arousal from hibernation in Arctic ground squirrel. Comparative Biochemistry and Physiology Part A: Molecular & Integrative Physiology. 2009; 153:213–21.
- Phifer-Rixey M, Bi K, Ferris KG, Sheehan MJ, Lin D, Mack KL, et al. The genomic basis of environmental adaptation in house mice. PLoS Genetics. 2018; 14:e1007672. https://doi.org/10.1371/journal.pgen. 1007672 PMID: 30248095
- 64. Hammond TT, Palme R, Lacey EA. Contrasting stress responses of two co-occurring chipmunk species (*Tamias alpinus* and *T. speciosus*). General and Comparative Endocrinology. 2015; 211:114–22. https://doi.org/10.1016/j.ygcen.2014.11.013 PMID: 25461808
- Carlson SM, Cunningham CJ, Westley PAH. Evolutionary rescue in a changing world. Trends in Ecology & Evolution. 2014; 29:521–30.
- Mills LS, Bragina EV, Kumar AV, Zimova M, Lafferty DJR, Feltner J, et al. Winter color polymorphisms identify global hot spots for evolutionary rescue from climate change. Science. 2018; 359:1033–6. https://doi.org/10.1126/science.aan8097 PMID: 29449510
- 67. Meyer M, Kircher M. Illumina sequencing library preparation for highly multiplexed target capture and sequencing. Cold Spring Harbor Protocols. 2010;2010:pdb.prot5448.
- Wang Z, Gerstein M, Snyder M. RNA-Seq: a revolutionary tool for transcriptomics. Nature Reviews Genetics. 2009; 10:57–63. https://doi.org/10.1038/nrg2484 PMID: 19015660
- 69. Singhal S. De novo transcriptomic analyses for non-model organisms: an evaluation of methods across a multi-species data set. Molecular Ecology Resources. 2013; 13:403–16. <u>https://doi.org/10.1111/</u> 1755-0998.12077 PMID: 23414390
- 70. Bi K, Vanderpool D, Singhai S, Linderoth T, Moritz C, Good J. Transcriptome-based exon capture enables highly cost-effective comparative genomic data collection at moderate evolutionary scales. BMC Genomics. 2012; 13:403. https://doi.org/10.1186/1471-2164-13-403 PMID: 22900609

- Good JM, Vanderpool D, Keeble S, Bi K. Negligible nuclear introgression despite complete mitochondrial capture between two species of chipmunks. Evolution. 2015; 69:1961–72. https://doi.org/10.1111/ evo.12712 PMID: 26118639
- 72. Good JM, Hird S, Reid N, Demboski JR, Steppan SJ, Martin-Nims TR, et al. Ancient hybridization and mitochondrial capture between two species of chipmunks. Molecular Ecology. 2008; 17:1313–27. https://doi.org/10.1111/j.1365-294X.2007.03640.x PMID: 18302691
- Simpson JT, Wong K, Jackman SD, Schein JE, Jones SJM, Birol I. ABySS: A parallel assembler for short read sequence data. Genome Research. 2009; 19:1117–23. https://doi.org/10.1101/gr.089532. 108 PMID: 19251739
- 74. Kent WJ. BLAT—The BLAST-like alignment tool. Genome Research. 2002; 12:656–64. https://doi.org/ 10.1101/gr.229202 PMID: 11932250
- Li WZ, Godzik A. Cd-hit: a fast program for clustering and comparing large sets of protein or nucleotide sequences. Bioinformatics. 2006; 22:1658–9. <u>https://doi.org/10.1093/bioinformatics/btl158</u> PMID: 16731699
- 76. Huang XQ, Madan A. CAP3: A DNA sequence assembly program. Genome Research. 1999; 9:868– 77. PMID: 10508846
- 77. Stiller M, Green RE, Ronan M, Simons JF, Du L, He W, et al. Patterns of nucleotide misincorporations during enzymatic amplification and direct large-scale sequencing of ancient DNA. Proceedings of the National Academy of Sciences of the United States of America. 2006; 103:13578–84. <u>https://doi.org/10.1073/pnas.0605327103</u> PMID: 16938852
- Korneliussen TS, Albrechtsen A, Nielsen R. ANGSD: analysis of next generation sequencing data. BMC Bioinformatics. 2014; 15:356. https://doi.org/10.1186/s12859-014-0356-4 PMID: 25420514
- 79. Fumagalli M, Vieira FG, Korneliussen TS, Linderoth T, Huerta-Sanchez E, Albrechtsen A, et al. Quantifying population genetic differentiation from next-generation sequencing data. Genetics. 2013; 195:979–92. https://doi.org/10.1534/genetics.113.154740 PMID: 23979584
- Fumagalli M, Vieira FG, Linderoth T, Nielsen R. ngsTools: methods for population genetics analyses from next-generation sequencing data. Bioinformatics. 2014; 30:1486–7. <u>https://doi.org/10.1093/</u> bioinformatics/btu041 PMID: 24458950
- Evanno G, Regnaut S, Goudet J. Detecting the number of clusters of individuals using the software STRUCTURE: a simulation study. Molecular Ecology. 2005; 14:2611–20. <u>https://doi.org/10.1111/j.</u> 1365-294X.2005.02553.x PMID: 15969739
- Alexander DH, Novembre J, Lange K. Fast model-based estimation of ancestry in unrelated individuals. Genome Research. 2009; 19:1655–64. https://doi.org/10.1101/gr.094052.109 PMID: 19648217
- Excoffier L, Foll M. fastsimcoal: a continuous-time coalescent simulator of genomic diversity under arbitrarily complex evolutionary scenarios. Bioinformatics. 2011; 27:1332–4. <u>https://doi.org/10.1093/bioinformatics/btr124</u> PMID: 21398675
- Csilléry K, Francois O, Blum MGB. abc: an R package for approximate Bayesian computation (ABC). Methods in Ecology and Evolution. 2012; 3:475–9.